Author’s response to reviews

Title: Prevalence of human papillomavirus infection among women from quilombo communities in Northeastern Brazil

Authors:

Maria do Desterro Nascimento (cnsd_ma@uol.com.br)
Flávia Castello Vidal (flavidaly@yahoo.com.br)
Marcos Antonio da Silva (marcosantonio456@hotmail.com)
José Eduardo Batista (jbatistaufma@gmail.com)
Maria do Carmo Barbosa (carminha13032009@hotmail.com)
Walbert Edson Muniz Filho (walbert.muniz@hotmail.com)
Geusa Felipa Bezerra (geusabezerra@yahoo.com.br)
Graça Maria Viana (gracaviana@globo.com)
Rebeca Costa Branco (bebecacastelo@hotmail.com)
Luciane Maria Brito (luciane2406@yahoo.com.br)

Version: 2 Date: 11 Dec 2017

Author’s response to reviews:

Dear Editor-in-chief

We would like to thank you and the reviewers for the commentaries about this manuscript.

In response to reviewer 1 comments

1. We added 1 paragraph about the justification for doing this research (Background section, page 5, lines 29-34)

2. We have reduced paragraphs regarding the description of HPV analysis methods.

3. We have reduced 1 table, as suggested.
Best regards,

Maria do Desterro Soares Brandão Nascimento.